Coronary Artery Disease

Consideration of a New Definition of Clinically Relevant Myocardial Infarction After Coronary Revascularization

An Expert Consensus Document From the Society for Cardiovascular Angiography and Interventions (SCAI)

Issam D. Moussa, MD,* Lloyd W. Klein, MD,† Binita Shah, MD,‡ Roxana Mehran, MD,§ Michael J. Mack, MD,|| Emmanouil S. Brilakis, MD,¶ John P. Reilly, MD,# Gilbert Zoghbi, MD,** Elizabeth Holper, MD,†† Gregg W. Stone, MD‡‡

Jacksonville, Florida; Chicago, Illinois; New York, New York; Dallas, Texas; New Orleans, Louisiana; and Memphis, Tennessee

Numerous definitions have been proposed for the diagnosis of myocardial infarction (MI) after coronary revascularization. The universal definition for MI designates post procedural biomarker thresholds for defining percutaneous coronary intervention (PCI)-related MI (type 4a) and coronary artery bypass grafting (CABG)-related MI (type 5), which are of uncertain prognostic importance. In addition, for both the MI types, cTn is recommended as the biomarker of choice, the prognostic significance of which is less well validated than CK-MB. Widespread adoption of a MI definition not clearly linked to subsequent adverse events such as mortality or heart failure may have serious consequences for the appropriate assessment of devices and therapies, may affect clinical care pathways, and may result in misinterpretation of physician competence. Rather than using an MI definition sensitive for small degrees of myonecrosis (the occurrence of which, based on contemporary large-scale studies, are unlikely to have important clinical consequences), it is instead recommended that a threshold level of biomarker elevation which has been strongly linked to subsequent adverse events in clinical studies be used to define a "clinically relevant MI." The present document introduces a new definition for "clinically relevant MI" after coronary revascularization (PCI or CABG), which is applicable for use in clinical trials, patient care, and quality outcomes assessment. (J Am Coll Cardiol 2013;62:1563–70) © 2013 by the American College of Cardiology Foundation

Numerous definitions for the diagnosis of MI after coronary revascularization are in use (1). A standardized MI definition would provide uniformity for comparing clinical trial results,

From the *Mayo Clinic, Jacksonville, Florida; †Rush University, Division of Cardiology, Department of Medicine, Chicago, Illinois; ‡New York University School of Medicine, New York, New York; §Mount Sinai Medical Center, Division of Cardiology, New York, New York; ||Baylor HealthCare System, Dallas, Texas; ¶VA North Texas Healthcare System and UT Southwestern Medical School, Dallas, Texas; #Oschner Clinic, New Orleans, Louisiana; **Stern Cardiovascular Foundation, Memphis, Tennessee; ††Medical City Dallas Hospital, Dallas, Texas; and the ‡‡Columbia University Medical Center, New York Presbyterian Hospital and the Cardiovascular Research Foundation, New York City, New York. Dr. Shah has received a research grant from Guerbet. Dr. Mehran is a consultant for The Medicines Company, BMS/Sanofi, Merck, Maya Medical, AstraZeneca, Merck, Abbott Vascular, Regado Biosciences, and Janssen Pharma; and has received grants or research support from Lilly/Daiichi Sankyo, The Medicines Company, and BMS/Sanofi. Dr. Brilakis is on the speakers' bureau for St. Jude Medical, Medtronic, and Terumo; has received research grants from Guerbet; has received honoraria from St. Jude Medical, Terumo, and Bridgepoint/Boston Scientific; and his spouse is an employee of Medtronic. Dr. Reilly is a consultant for Gilead; and is on the speakers' bureau for AstraZeneca and Eli Lilly/DSI. Dr. Holper is on the speakers' bureau for Boston Scientific. Dr. Stone is a consultant for Boston Scientific, Eli Lilly, Daiichi Sankyo, and AstraZeneca. All other authors have reported that they have no relationships relevant to the contents of this paper to disclose.

Manuscript received July 12, 2013; accepted July 13, 2013.

for assessing patient outcomes and for guiding quality improvement initiatives. In 2007, a "universal definition" for MI following coronary revascularization was proposed (2) and recently revised in 2012 (3). In this document, a PCI-related MI (type 4a) was defined as an increase in cTn to >5× the 99th percentile of the URL during the first 48 h following PCI (in patients with normal baseline cTn concentrations), plus

See page 1571

either: 1) evidence of prolonged ischemia as demonstrated by prolonged chest pain; or 2) ischemic ST-segment changes or new pathological Q waves; or 3) angiographic evidence of a flow limiting complication; or 4) imaging evidence of new loss of viable myocardium or new regional wall motion abnormality. MI associated with CABG (type 5) was defined as an increase in cTn to >10× the 99th percentile URL during the first 48 h following CABG (in patients with normal baseline cTn concentrations), plus either: 1) new pathological Q waves or new LBBB; or 2) angiographically documented

Abbreviations and Acronyms

ACS = acute coronary syndrome(s)

CABG = coronary artery bypass graft

cMRI = cardiac magnetic resonance imaging

LGE = late gadolinium enhancement

MI = myocardial infarction

PCI = percutaneous coronary intervention new graft or new native coronary artery occlusion; or 3) imaging evidence of new loss of viable myocardium or new regional wall motion abnormality. cTn were considered preferred the biomarker for detection of myonecrosis. However, the writing committee also noted that these definitions were arbitrarily chosen and of uncertain clinical relevance, and not grounded on substantial scientific evidence linking their occurrence to subse-

quent adverse outcomes (2,3).

Assessment of post-PCI and -CABG biomarkers that are strongly related to subsequent adverse patient outcomes is undoubtedly worthwhile. Conversely, applying undue significance to peri-procedural biomarker elevations without prognostic relevance may have unintended consequences on patient care, physician and systems evaluation of PCI quality, and for the development and appropriate assessment of new therapies. Uncertainty among healthcare providers regarding the interpretation of elevated cardiac biomarkers after successful revascularization can lead to prolonged hospital stay and unnecessary diagnostic or therapeutic interventions which may result in iatrogenic complications and increased costs. Lastly, adoption of a MI definition not based on meaningful correlation with adverse consequences in clinical trials may result in false conclusions regarding the relative risk-benefit ratio of comparative therapeutic strategies. The present working group was thus tasked by SCAI Publication committee with developing a definition of periprocedural MI which, based on the preponderance of the best scientific evidence, has clearly been shown to have important prognostic significance, hence termed a "clinically relevant" MI. This document is endorsed by SCAI publication committee, SCAI board of trustees, and SCAI executive committee.

The present document will emphasize the definition of MI after PCI procedures, but also refer to MI after CABG. The present consensus recommendations are based heavily on data from the largest, contemporary studies. The reader should view this document as the best attempt of the writing committee to inform and guide clinical practice and clinical trial design in an area where the evidence base, while substantial, is evolving.

Pathophysiology of Peri-Procedural Myonecrosis, and the Potential for Confounding

The pathophysiology of myocardial necrosis following revascularization procedures is multifactorial (4–8) (Online Table 1). The mechanisms and risk factors associated with post-PCI myonecrosis vary according to the definition of myonecrosis (e.g., troponin vs. CK-MB elevations, and the

selected threshold for abnormality) and the patient population studied. Lee et al. (6) reported that a post-PCI MI (defined as cTn I elevation to $>3 \times$ URL) was predicted by treatment of type B2/C lesions and a thin-cap fibroatheroma as assessed by optical coherence tomography. Other studies have shown a strong association between post-procedural cardiac biomarker release and large atherosclerotic plaque burden, large thrombus burden, coronary calcification, and lesion eccentricity, as detected by angiography and intravascular ultrasound imaging (7-9). As the extent and complexity of coronary atherosclerosis is also an independent predictor of mortality after PCI (10,11), the association between post-PCI biomarker elevation and mortality may be an epiphenomenon; that is, the linkage is explained by the impact of the extent of atherosclerosis and lesion complexity on mortality, not the biomarker elevation itself (12). Thus, multivariable analysis incorporating both clinical and angiographic variables, along with biomarker elevations is necessary to determine whether the biomarker elevation per se is an independent correlate of mortality.

Angiographically evident complications are not always associated with sizable post-PCI biomarker elevations, and biomarker elevations can occur without angiographic complications (5,13). Muschart et al. (5) identified an angiographic cause of post-PCI CK-MB >ULN (sidebranch occlusion, distal embolization, slow flow or no-reflow, intraprocedural stent thrombosis, or coronary perforation) in only 60% of cases. Whether periprocedural biomarker elevations of any level correlate with subsequent adverse events when angiographic complications are absent is questionable. Among 5,850 patients from six stent trials, even large biomarker elevations (CK-MB >8× ULN) had no prognostic impact unless associated with overt angiographic complications, questioning the relevance of isolated measures of myonecrosis (14). The frequency of peri-procedural MI depends on the type of cardiac biomarker being measured (cTn vs. CK-MB), the frequency and timing of biomarker determinations, and the threshold that defines a MI (15). Moreover, in most clinical studies, the upper range of cardiac biomarkers has been defined as a multiple of the ULN of the local laboratory assay. In contrast, the universal definition recommends evaluating biomarker elevations based on the 99th percentile limit of the reference value distribution (99th upper reference limit [URL]) (2,3), which is an even more sensitive measure. However, use of the URL is problematic because most assays do not have the sensitivity to consistently measure cTn in the blood of apparently healthy individuals, with a high proportion of the values being below the method's detection limit. Moreover, the 99th URL for cTn can vary depending on the reference population used, and its sample size. A marked difference can also be observed between the 99th URL obtained using plasma versus that obtained with serum. Finally, interferences with cTn measurement may occur with assays having increased analytical sensitivity (16). As a result of these limitations, most laboratories have established a normal reference range defined as the set of values within which 95% of the normal population falls. Most prior studies have used this locally derived ULN when examining the prognostic relevance of CK-MB and troponin elevations.

Importance of Pre-Procedure Cardiac Biomarker Elevation

Prior studies in which pre-PCI and post-PCI cTn elevations were analyzed have reported that only pre-procedure cTn elevations are correlated with subsequent mortality (Online Table 2). In an analysis of 5,487 patients undergoing non-emergent PCI at the Mayo Clinic (17), an abnormal pre-PCI cTnT level independently predicted death (median follow-up of 28 months), but the occurrence of PCI-related myonecrosis did not, whether defined by >3× elevation in cTn or CK-MB. In a separate study, baseline cTn was >ULN in 6.0% of 2,382 patients undergoing elective PCI, and was an independent predictor of in-hospital death or MI (18). Thus, interpretation of post-PCI biomarker elevations may be erroneous if baseline levels are not assessed.

The accurate diagnosis of post-PCI MI is challenging among patients with baseline biomarker elevations. The 2012 universal definition states that "If the baseline cTn values are elevated and are stable or falling, then a rise of >20% is required for the diagnosis of a type 4a MI, as with reinfarction" (3). These recommendations were based on recent data from the TRITON-TIMI-38 which suggest that, when PCI is delayed after MI until biomarker concentrations are falling or have normalized, re-elevation of cardiac biomarker values may have long-term significance (19). The diagnosis of peri-PCI MI in this study, however, was preferentially based on CK-MB measures, and residual confounding cannot be excluded, as discussed in the next section. As such, the prognostic relevance of post-PCI biomarkers in the patient with a recent ACS have not been sufficiently validated to be applied clinically.

cTn Versus CK-MB Elevation After Coronary Revascularization

cTn is a more sensitive and specific biomarker for myonecrosis than is CK-MB (20,21), and CK-MB is no longer measured at some institutions. This transition was made prior to thorough understanding of the prognostic implications of cTn elevations after coronary revascularization. Numerous studies have suggested a stronger association with post-PCI CK-MB and subsequent cardiovascular events than with cTn elevation (Online Table 3). Furthermore, the greater sensitivity of cTn for myonecrosis markedly increases the rate of type 4a MI. Among 4,930 patients undergoing elective coronary stenting, MI occurred in 7.2% using a CK-MB criteria of \geq 3× ULN and in 24.3% using a cTn criteria of \geq 3× ULN (22). Lastly, changes in myocycte membrane permeability resulting from prolonged myocardial ischemia and other causes is sufficient for the release of cTn from the

free cytosolic pool of myocytes without structural damage, explaining troponin elevations during rapid atrial pacing, supraventricular tachycardia and noncardiac conditions, including exercise and renal disease (23).

The recommendation to use cTn rather than CK-MB and a cutoff of 3× the 99th percentile URL for the diagnosis of post-PCI MI in the original universal definition of MI was described as arbitrary by the authors (2). The revised universal definition has raised the cTn cutoff to 5× the 99th percentile URL and added the requirement for clinical, ECG or angiographic changes for the diagnosis of post-PCI MI, likely increasing the specificity for MI (3). Although these revisions are a step in the right direction, they are still admittedly arbitrary and do not sufficiently weigh the contemporary evidence base from which the criteria for a "clinically relevant MI" (one associated with an adverse prognosis) should be derived.

The relationship between biomarker elevations after revascularization and mortality is controversial, and the evidence base for this association has evolved over the last 15 years. When analyzed as a continuous variable above a threshold, a small incremental risk with cardiac biomarker elevation is noted (Online Table 3). However, this type of analysis is misleading because it lumps large biomarker elevations with small elevations. When analyzed in categories of incrementally increasing biomarker elevations, most contemporary PCI studies have reported associations between peri-procedural myonecrosis and mortality only for very large infarcts (Online Table 3). Among 7,147 patients undergoing PCI at Washington Hospital Center, 37.3% had a post-PCI CK-MB >ULN; however, only Q-wave MI or non-Q-wave MI with a peak CK-MB of $>8 \times$ ULN correlated with increased 2-year mortality (24). Among 3,478 patients undergoing stent implantation at the Cleveland Clinic, post-PCI CK-MB >ULN was detected in 24%; only CK-MB >10× ULN was significantly associated with increased 1-year mortality (25). In the EVENT registry (n = 6,347), only CK-MB \geq 50 mg/ml ($\sim 10 \times ULN$) was associated with increased 30-day and 1-year mortality after nonemergent PCI (26). In the 3,687 patient SPIRIT IV trial, CK-MB >3× ULN and cTn >3× ULN were present in 5.4% and 19.7% of stented patients, respectively; there was no relationship between any level of either biomarker and 2-year mortality, even >10× ULN (27). Some have argued that small peri-procedural MIs will only have clinical impact with long-term follow-up. However, there was no relationship between protocol-defined MI and 5-year mortality among 5,467 patients with ACS in whom PCI was performed from three randomized trials (28).

The TRITON-TIMI-38 investigators did report a relationship between type 4a MI (defined as CK-MB $>3\times$ URL on two samples or $>5\times$ on a single sample) and 6-month cardiovascular mortality (19). However, this risk was not adjusted for the complexity of underlying CAD or treated lesions. Similarly, multivariable analysis to account for recent MI, unstable angina, thrombotic or complex

lesions, vein graft intervention, and various device usages was not performed in many of the earlier studies examining the relationship between peri-procedural biomarker elevation and mortality. As previously discussed, mortality after CK-MB elevation may reflect the underlying extent of atherosclerosis and lesion complexity requiring treatment rather than the biomarker elevation per se (12). Similarly, procedural urgency and multivessel PCI is more common in patients with small post-PCI cTn elevations, suggesting complex and extensive atherosclerosis as underlying the mortality associated with microinfarcts rather than minor myocardial injury (29). The association of low-level biomarker elevations with mortality in early studies was always difficult to reconcile with traditional concepts relating large infarctions and depressed global cardiac function as major determinants of prognosis. Conversely, nearly all large-scale, contemporary studies have shown that only a substantial amount of myonecrosis, typically associated with CK-MB >10× ULN or new Q-wave infarction, is required to adversely impact survival.

The requirement for a large biomarker increase to signify a clinically relevant MI after PCI contrast with type I (spontaneous) MI, in which even minor elevations of cTn have repeatedly and strongly been associated with high rates of death and subsequent MI during follow-up (17). The adverse prognosis following type I MI is not only due to the immediate impact of myocardial injury, but also reflects the consequences of plaque rupture occurring in an uncontrolled setting. Multifocal plaque instability and persistent systemic inflammation are also often present and increase the risk of future events (30–32). Conversely, peri-procedural myonecrosis is induced by a single procedure in a controlled setting, and in most cases its consequences (if any) are immediately addressed.

Insight From Cardiac Magnetic Resonance Imaging Studies

cMRI is increasingly being used to evaluate the extent of coronary hypoperfusion and myonecrosis (33,34). Large compared to small or absent cMRI defects are strongly associated with subsequent mortality (33). cMRI studies have demonstrated poor discrimination of cTn elevations for myocardial injury. Using late gadolinium enhancement (LGE), Lim et al. (35) reported the sensitivity, specificity and positive predictive value of cTn I-defined MI $>3\times$ the 99th percentile URL to be 100, 22, and 19%, respectively. The analogous findings using CK-MB-defined MI $>3\times$ the 99th percentile URL were 60, 93, and 60%, respectively, demonstrating that cTn is overly sensitive and identifies many patients without objective evidence of cMRI-detected myocardial injury. While small levels of cTn elevation may represent myonecrosis not detectable by cMRI, the prognostic significance of such small elevations has not been demonstrated. The authors further reported that changing the cTn I threshold for MI diagnosis to 40× the

99th percentile URL would greatly enhance specificity (93%) without reducing sensitivity (100%). This ~13× bioequivalence ratio of cTn:CK-MB is even greater than the ~7-fold ratio reported from the EVENT registry in which a cTn threshold >20× ULN was associated with similar mortality rates as a >3× CK-MB increase (22). These data confirm that very different thresholds for CK-MB and cTn after PCI must be considered to represent similar levels of myonecrosis.

Recommendation for a Definition of "Clinically Relevant MI" After Coronary Revascularization

Patients with normal baseline cardiac biomarkers. Compilation of the best medical evidence to date does not support use of the universal definition as the optimal criterion to identify clinically relevant post-PCI MI events. Rather, most contemporary studies support a post-PCI elevation of CK-MB to $\geq 10 \times$ ULN as being clinically relevant. While there may be some incremental value of identifying very small post-PCI infarcts, specifically those that occur with angiographically evident complications, more study is required in this regard, with angiographic core laboratory verification of events. Similarly, almost all clinical studies to date have reported the impact of biomarker elevation in relation to a multiple of the ULN, not the 99th percentile of the URL. Thus, absent such studies, a clinically relevant MI occurring in the post-PCI period should be defined as that resulting in a CK-MB $\geq 10 \times$ ULN. A lower threshold $(\geq 5 \times \text{ULN})$ may be accepted in the patient in whom new pathologic Q-waves in ≥2 contiguous leads (or new persistent LBBB) develop post-PCI, although further study is required to validate this threshold in the setting of new Q-

A threshold post-PCI level of cTn above which long-term prognosis is affected has not been established, and thus CK-MB is strongly preferred to assess clinically relevant post-PCI MI events. If CK-MB levels are unavailable, a reasonable cTn (I or T) value to substitute to diagnose a type 4a MI would be cTn of \geq 70× ULN, based on the 7:1 troponin:CK-MB ratio noted to have approximate similar clinical implications (22). As discussed, this 7:1 ratio is a conservative estimate when examining the results from imaging studies (35). Thus, in patients with normal baseline cTn and without an acute coronary syndrome (ACS) (i.e., rising biomarker levels are not suspected), a clinically relevant MI post-PCI is diagnosed by a new biomarker elevation of CK-MB to $\geq 10 \times$ ULN or cTn (I or T) to $\geq 70 \times$ ULN (or by CK-MB to $\geq 5 \times$ ULN or cTn to $\geq 35 \times$ ULN plus the development of new pathologic Q-waves in ≥ 2 contiguous leads or LBBB).

Patients with elevated baseline cardiac biomarkers. Accurately diagnosing post-PCI MI in the setting of elevated baseline biomarkers is problematic, and depends on whether the peak level has been documented. This requires assessment of serial biomarker levels. Often, however,

cardiac catheterization is performed after only a single biomarker level is obtained, or when serial biomarkers show continued escalation, in which case diagnosing post-PCI MI on the basis of biomarker levels alone is impossible. Thus, based on the thresholds determined from large-scale studies in which baseline biomarkers were normal, the following recommendations are made to diagnose post-PCI MI in ACS patients in whom the baseline level has not returned to normal: 1) In patients with elevated cTn (or CK-MB) in whom the biomarker levels are stable or falling, there should be a new CK-MB elevation by an absolute increment of $\geq 10 \times$ ULN (or $\geq 70 \times$ ULN for cTn I or T) from the previous nadir level; 2) In patients with elevated cTn (or CK-MB) in whom the biomarker levels have not been shown to be stable or falling, there should be a further rise in CK-MB or troponin beyond the most recently measured value by an absolute increment of $\geq 10 \times$ ULN in CK-MB or $\geq 70 \times$ ULN in cTn plus new ST-segment elevation or depression plus signs consistent with a clinically relevant MI, such as new onset or worsening heart failure or sustained hypotension. Chest pain alone is not specific enough for substantial myonecrosis to be used as a criterion. We acknowledge, however, that further studies are required to refine the best diagnostic criteria in patients with elevated baseline biomarkers.

These criteria are summarized in Table 1. Practical recommendations of a clinically relevant definition of MI for assessment of cardiac biomarkers after PCI, its use as a quality of care indicator, and its impact on clinical trial design are discussed in the Online Appendix.

Post-CABG Myocardial Infarction

Applying the 2007 universal definition of post CABG MI (type 5), 42% to 82% of patients have cardiac biomarker elevation ≥5× the URL (36,37), but only 4% to 7% have the additional electrocardiographic evidence needed to meet the definition of post-CABG MI (38,39). In the 2012 revised version of the universal definition the threshold for diagnosis of post CABG MI (type 5) was increased to >10× the 99th percentile of the URL from a normal baseline cTn value and in addition requires either: 1) new pathological Q-waves or new LBBB; or 2) angiographically documented new graft or new native coronary artery occlusion; or 3 imaging evidence

of new loss of viable myocardium or new regional wall motion abnormality (3). The implications of this change on the frequency and prognostic implications of post-CABG MI are unclear.

Although some studies have suggested that only large myocardial biomarker elevations after CABG are clinically important (40-42), others have demonstrated that even modest CK-MB elevations after CABG may be associated with increased mortality (43-48). Most recently, a large patientlevel meta-analysis demonstrated a doubling of short- and long-term mortality in patients in whom cardiac biomarker measurements within 24 hr after CABG rose to ≥ 4 to $5 \times$ ULN (49). After multivariable adjustment, 30-day mortality was significantly increased in patients with a post-CABG CK-MB of $>5 \times$ ULN and with a cTn $>40 \times$ ULN; an exponential rise in mortality was seen with $>40\times$ and $>100\times$ increases in CK-MB and cTn above the ULN, respectively. Additional studies have suggested that early biomarker release (occurring within 1 h post-CABG) may be a meaningful predictor of in-hospital mortality (50), while elevation in later samples (with 48 h post-CABG) appear to be a stronger predictor of long-term mortality (51).

Modest-sized studies utilizing cMRI suggest that 28% to 44% of patients post-CABG have detectable myonecrosis (50–52). These studies have shown a moderate correlation between post-CABG cardiac biomarker elevation and the extent of new myocardial necrosis observed by LGE. Moreover, these studies have demonstrated that new areas of LGE after CABG have greater prognostic impact when accompanied by biomarker elevation of $\geq 5 \times$ ULN compared to $< 5 \times$ ULN.

Similar to the post-PCI population, the majority of the studies have evaluated the prognostic significance of post-CABG cardiac biomarker elevations using CK-MB. cTn measurements may be an overly sensitive marker of myocardial necrosis in this setting. Remmpis et al. have shown that cTn release from nonstructurally bound cytosolic pools may lead to systemic troponin elevations without cardiomyocyte injury (53, Online Refs. 54–99). "Washout" of this cytosolic component of troponin, secondary to surgery-induced increased cell permeability without cellular death, could result in cTn rise in the first few hours after surgery without irreversible myocardial injury.

Table 1 **Definition of Clinically Relevant MI After Both PCI and CABG Procedures** In patients with normal baseline CK-MB The peak CK-MB measured within 48 hours of the procedure rises to \geq 10 \times the local laboratory ULN, or to $\geq\!\!5\times$ ULN with new pathologic Q-waves in $\geq\!\!2$ contiguous leads or new persistent LBBB, OR in the absence of CK-MB measurements and a normal baseline cTn, a cTn (I or T) level measured within 48 hours of the PCI rises to >70 \times the local laboratory ULN, or >35 \times ULN with new pathologic Q-waves in ≥2 contiguous leads or new persistent LBBB. The CK-MB (or cTn) rises by an absolute increment equal to those levels recommended above from In patients with elevated baseline CK-MB (or cTn) in whom the biomarker levels are stable or falling the most recent pre-procedure level. In patients with elevated CK-MB (or cTn) in whom the biomarker The CK-MB (or cTn) rises by an absolute increment equal to those levels recommended above plus levels have not been shown to be stable or falling new ST-segment elevation or depression plus signs consistent with a clinically relevant MI, such as new onset or worsening heart failure or sustained hypotension.

The rationale for requiring a $\geq 10 \times$ increase in cardiac biomarkers for CABG versus a >5× increase for PCI as recommended in the 2012 universal definition is not clearly substantiated (3). Nonetheless, as a working definition we support this threshold to diagnose a clinically relevant MI post CABG. However, CK-MB is the preferred biomarker, and if a cTn threshold must be used, $\geq 70 \times$ is reasonable. Thus, the recommendations for the preferred biomarker type and threshold after PCI and CABG are harmonized (Table 1). While not currently recommended as part of this definition (given the absence of data), the use of post-CABG ECGs, indices of hemodynamic instability, and imaging studies demonstrating new wall motion abnormalities have been suggested to complement biomarker elevations post-CABG to improve specificity. Additional studies in this regard are warranted.

Limitations and Future Directions

The currently recommended definition of a clinically relevant MI, while admittedly imperfect, is based on the best scientific evidence presently available. Nonetheless, we acknowledge the need for further research, and revision of this definition as new data becomes available. In particular, given the greater specificity of cTn for cardiac myonecrosis compared to CK-MB, additional investigation should focus on determining the threshold at which cTn measurements have prognostic value after PCI and CABG (and whether in this regard there are important differences between cTnT and cTnI).

One of the greatest barriers in interpreting the current data is the lack of patient-level, vessel-level and lesion-level characteristics, making it difficult to attribute post-PCI biomarker elevations to adverse events rather than representing a surrogate marker of higher risk patients. Future research should consider the relationship between plaque burden, coronary anatomy and complex intervention on the incidence of peri-procedural MI. Until the issues surrounding diagnosis and prognosis associated with periprocedural myonecrosis and adverse events can be resolved, our ability to effectively risk-stratify and optimally manage patients undergoing coronary revascularization will remain limited, and evaluation of the utility and cost-effectiveness of new therapies will be imperfect. It is our belief that the use of the definition of clinically relevant MI introduced in this document is a first step toward addressing these issues.

Reprint requests and correspondence: Dr. Issam D. Moussa, Division of Cardiovascular Diseases, Mayo Clinic, 4500 San Pablo Road, Jacksonville, Florida 32224. E-mail: moussa.issam@mayo.edu.

REFERENCES

1. Leonardi S, Armstrong PW, Schulte PJ, et al. Implementation of standardized assessment and reporting of myocardial infarction in

- contemporary randomized controlled trials: A systematic review. Eur Heart J 2013;34:894–902.
- Thygesen K, Alpert JS, White HD, et al. Universal definition of myocardial infarction. Circulation 2007;116:2634–53.
- Thygesen K, Alpert JS, Allan S, et al. Third universal definition of myocardial infarction. J Am Coll Cardiol 2012;60:1581–98.
- Lim CC, Cuculi F, van Gaal WJ, et al. Early diagnosis of perioperative myocardial infarction after coronary bypass grafting: A study using biomarkers and cardiac magnetic resonance imaging. Ann Thorac Surg 2011;92:2046–53.
- Muschart X, Slimani A, Jamart J, et al. The different mechanisms of periprocedural myocardial infarction and their impact on in-hospital outcome. J Invasive Cardiol 2012;24:655–60.
- Lee T, Yonetsu T, Koura K, et al. Impact of coronary plaque morphology assessed by optical coherence tomography on cardiac troponin elevation in patients with elective stent implantation. Circ Cardiovasc Interv 2011;4:378–86.
- Mehran R, Dangas G, Mintz GS, et al. Atherosclerotic plaque burden and CK-MB enzyme elevation after coronary intervention: Intravascular ultrasound study of 2256 patients. Circulation 2000;101:604–10.
- Kanaparti PK, Brown DL. Relation between coronary atherosclerotic plaque burden and cardiac enzyme elevation following percutaneous coronary intervention. Am J Cardiol 2000;86:619–22.
- van Gaal WJ, Ponnuthurai FA, Selvanayagam J, et al. The syntax score predicts peri-procedural myocardial necrosis during percutaneous coronary intervention. Int J Cardiol 2009;135:60–5.
- Garg S, Sarno G, Girasis C, et al. A patient-level pooled analysis assessing the impact of the SYNTAX (synergy between percutaneous coronary intervention with taxus and cardiac surgery) score on 1-year clinical outcomes in 6,508 patients enrolled in contemporary coronary stent trials. J Am Coll Cardiol Intv 2011;4:645-53.
- Palmerini T, Genereux P, Caixeta A, et al. Prognostic value of the SYNTAX score in patients with acute coronary syndromes undergoing percutaneous coronary intervention: Analysis from the ACUITY (acute catheterization and urgent intervention triage strategy) trial. J Am Coll Cardiol 2011;57:2389–97.
- Cutlip DE, Kuntz RE. Cardiac enzyme elevation after successful percutaneous coronary intervention is not an independent predictor of adverse outcomes. Circulation 2005;112:916–23.
- 13. Blankenship JC, Islam MA, Wood CG, Iliadis EA. Angiographic adverse events during percutaneous coronary intervention fail to predict creatine kinase-MB elevation. Cath Cardiovasc Int 2004;63:31–41.
- Jeremias A, Baim DS, Ho KK, et al. Differential mortality risk of postprocedural creatine kinase-MB elevation following successful versus unsuccessful stent procedures. J Am Coll Cardiol 2004;44:1210–4.
- Lansky AJ, Stone GW. Periprocedural myocardial infarction. Prevalence, prognosis, and prevention. Circ Cardiovasc Interv 2010;3:602–10.
- Panteghini M. A critical appraisal of experimental factors influencing the definition of the 99th percentile limit for cardiac troponins. Clin Chem Lab Med 2009;47:1179–82.
- 17. Prasad A, Rihal CS, Lennon RJ, et al. Significance of periprocedural myonecrosis on outcomes after percutaneous coronary intervention: An analysis of preintervention and postintervention troponin T levels in 5487 patients. Circ Cardiovasc Interv 2008;1:10–9.
- 18. Jeremias A, Kleiman NS, Nassif D, et al., Evaluation of drug eluting stents and ischemic events (EVENT) registry investigators. Prevalence and prognostic significance of preprocedural cardiac troponin elevation among patients with stable coronary artery disease undergoing percutaneous coronary intervention: Results from the evaluation of drug eluting stents and ischemic events registry. Circulation 2008;118:632–8.
- Bonaca MP, Wiviott SD, Braunwald E, et al. Universal definition of myocardial infarction classification system and the risk of cardiovascular death: Observations from the TRITON-TIMI 38 trial (Trial to Assess Improvement in Therapeutic Outcomes by Optimizing Platelet Inhibition With Prasugrel-Thrombolysis In Myocardial Infarction 38). Circulation 2012;125:577-83.
- 20. Jaffe AS, Ravkilde J, Roberts R, et al. It's time for a change to a troponin standard. Circulation 2000;102:1216–20.
- Adams JE III, Bodor GS, Davila-Roman VG, et al. Cardiac troponin I. A marker with high specificity for cardiac injury. Circulation 1993; 88:101–6.
- Novack V, Pencina M, Cohen DJ, et al. Troponin criteria for myocardial infarction after percutaneous coronary intervention. Arch Intern Med 2012;172:502–8.

- Hamm CW, Giannitsis E, Katus HA. Cardiac troponin elevations in patients without acute coronary syndrome. Circulation 2002;106: 666–71.
- 24. Stone GW, Mehran R, Dangas G, et al. Differential impact on survival of electrocardiographic Q-wave versus enzymatic myocardial infarction after percutaneous intervention: A device-specific analysis of 7147 patients. Circulation 2001;104:642–7.
- Brener SJ, Ellis SG, Schneider J, Topol EJ. Frequency and long-term impact of myonecrosis after coronary stenting. Eur Heart J 2002;23: 869–76.
- 26. Lindsey JB, Kennedy KF, Stolker JM, et al. Prognostic implications of creatine kinase-MB elevation after percutaneous coronary intervention: Results from the evaluation of drug-eluting stents and ischemic events (EVENT) registry. Circ Cardiovasc Interv 2011;4:474–80.
- Stone GW, Řizvi A, Newman W, et al., SPIRIT IV investigators. Everolimus-eluting versus paclitaxel-eluting stents in coronary artery disease. N Engl J Med 2010;362:1663–74.
- 28. Damman P, Wallentin L, Fox KA, et al. Long-term cardiovascular mortality after procedure-related or spontaneous myocardial infarction in patients with non-ST-segment elevation acute coronary syndrome: A collaborative analysis of individual patient data from the FRISC II, ICTUS, and RITA-3 trials (FIR). Circulation 2012;125:568–76.
- Ricciardi MJ, Wu E, Davidson CJ, et al. Visualization of discrete microinfarction after percutaneous coronary intervention associated with mild creatine kinase-MB elevation. Circulation 2001;103: 2780–3.
- 30. Rioufol G, Finet G, Ginon I, et al. Multiple atherosclerotic plaque rupture in acute coronary syndrome: A three-vessel intravascular ultrasound study. Circulation 2002;106:804–8.
- Tanaka A, Shimada K, Sano T, et al. Multiple plaque rupture and C-reactive protein in acute myocardial infarction. J Am Coll Cardiol 2005; 45:1594–9.
- Zhao Z, Witzenbichler B, Mintz GS, et al. Dynamic nature of nonculprit coronary artery lesion morphology in STEMI. J Am Coll Cardiol Img 2013;6:86–95.
- Klem I, Shah DJ, White RD, et al. Prognostic value of routine cardiac magnetic resonance assessment of left ventricular ejection fraction and myocardial damage: An international, multicenter study. Circ Cardiovasc Imaging 2011;4:610–9.
- 34. Porto I, Selvanayagam JB, Van Gaal WJ, et al. Plaque volume and occurrence and location of periprocedural myocardial necrosis after percutaneous coronary intervention: Insights from delayedenhancement magnetic resonance imaging, thrombolysis in myocardial infarction myocardial perfusion grade analysis, and intravascular ultrasound. Circulation 2006;114:662–9.
- 35. Lim CC, van Gaal WJ, Testa L, et al. With the "universal definition," measurement of creatine kinase-myocardial band rather than troponin allows more accurate diagnosis of periprocedural necrosis and infarction after coronary intervention. J Am Coll Cardiol 2011;57:653–61.
- Eigel P, van Ingen G, Wagenpfeil S. Predictive value of perioperative cardiac troponin I for adverse outcome in coronary artery bypass surgery. Eur J Cardiothorac Surg 2001;20:544–9.
- 37. Selvanayagam JB, Petersen SE, Francis JM, et al. Effects of off-pump versus on-pump coronary surgery on reversible and irreversible myocardial injury: A randomized trial using cardiovascular magnetic resonance imaging and biochemical markers. Circulation 2004;109:345–50.
- 38. Mahaffey KW, Roe MT, Kilaru R, et al. Creatine kinase-MB elevation after coronary artery bypass grafting surgery in patients with non-ST-segment elevation acute coronary syndromes predict worse outcomes: Results from four large clinical trials. Eur Heart J 2007;28:425–32.
- Nesher N, Alghamdi AA, Singh SK, et al. Troponin after cardiac surgery: a predictor or a phenomenon? Ann Thorac Surg 2008;85: 1348-54.
- Paparella D, Cappabianca G, Visicchio G, et al. Cardiac troponin I release after coronary artery bypass grafting operation: Effects on operative and midterm survival. Ann Thorac Surg 2005;80:1758–64.
- Ramsay J, Shernan S, Fitch J, et al. Increased creatine kinase MB level predicts postoperative mortality after cardiac surgery independent of new Q waves. J Thorac Cardiovasc Surg 2005;129:300–6.
- 42. Klatte K, Chaitman BR, Theroux P, et al. Increased mortality after coronary artery bypass graft surgery is associated with increased levels of postoperative creatine kinase-myocardial band isoenzyme release: Results from the GUARDIAN trial. J Am Coll Cardiol 2001;38: 1070–7.

- 43. Costa MA, Carere RG, Lichtenstein SV, et al. Incidence, predictors, and significance of abnormal cardiac enzyme rise in patients treated with bypass surgery in the arterial revascularization therapies study (ARTS). Circulation 2001;104:2689–93.
- Croal BL, Hillis GS, Gibson PH, et al. Relationship between postoperative cardiac troponin I levels and outcome of cardiac surgery. Circulation 2006;114:1468–75.
- 45. Vikenes K, Andersen KS, Melberg T, et al. Long time prognostic value of CK-MB mass in low-risk patients with stable angina scheduled for cardiac surgery. Cardiology 2009;113:122–31.
- Lurati Buse GA, Koller MT, Grapow M, et al. The prognostic value of troponin release after adult cardiac surgery—A meta-analysis. Eur J Cardiothorac Surg 2010;37:399–406.
- van Geene Y, van Swieten HA, Noyez L. Cardiac troponin I levels after cardiac surgery as predictor for in-hospital mortality. J Thorac Cardiovasc Surg 2009;10:413–6.
- Petaja L, Salmenpera M, Pulkki K, et al. Biochemical injury markers and mortality after coronary artery bypass grafting: A systematic review. Ann Thorac Surg 2009;87:1981–92.
- Domanski MJ, Mahaffey K, Hasselblad V, et al. Association of myocardial enzyme elevation and survival following coronary artery bypass graft surgery. JAMA 2011;305:585–91.
- Van Gall WJ, Arnold JR, Testa L, et al. Myocardial injury following coronary artery surgery versus angioplasty (MICASA): A randomized trial using biochemical markers and cardiac magnetic resonance imaging. EuroIntervention 2011;6:703–10.
- Steuer J, Berner T, Duvernoy O, et al. Visualization and quantification of perioperative myocardial infarction after coronary artery bypass surgery with contrast-enhanced magnetic resonance imaging. Eur Heart J 2004;25:1293–9.
- 52. Rahimi K, Banning AP, Cheng ASH, et al. Prognostic value of coronary revascularization-related myocardial injury: A cardiac magnetic resonance imaging study. Heart 2009;95:1937–43.
- Remppis A, Scheffold T, Greten J, et al. Intracellular compartmentation of troponin T: Release kinetics after global ischemia and calcium paradox in the isolated perfused rat heart. J Mol Cell Cardiol 1995;27: 793–803.
- 54. Fitzmaurice TF, Brown C, Rifai N, et al. False increase of cardiac troponin I with heterophilic antibodies. Clin Chem 1998;44:2212-4.
- Krahn J, Parry DM, Leroux M, et al. High percentage of false positive cardiac troponin I results in patients with rheumatoid factor. Clin Biochem 1999;32:477–80.
- Bodor GS, Survant L, Voss EM, et al. Cardiac troponin T composition in normal and regenerating human skeletal muscle. Clin Chem 1997; 43:476–84.
- 57. Wang TY, Peterson ED, Dai D, et al. Patterns of cardiac marker surveillance after elective percutaneous coronary intervention and implications for the use of periprocedural myocardial infarction as a quality metric. J Am Coll Cardiol 2008;51:2068–74.
- 58. Levine GN, Bates ER, Blankenship JC, et al. 2011 ACCF/AHA/ SCAI guideline for percutaneous coronary intervention: a report of the American College of Cardiology Foundation/American Heart Association Task Force on Practice Guidelines and the Society for Cardiovascular Angiography and Interventions. J Am Coll Cardiol 2011;58:e44–122.
- Morrow DA. Clinical application of sensitive troponin assays. N Engl J Med 2009;361:913–5.
- 60. Venge P, Johnston N, Lindahl B, et al. Normal plasma levels of cardiac troponin I measured by the high-sensitivity cardiac troponin I access prototype assay and the impact on the diagnosis of myocardial ischemia. J Am Coll Cardiol 2009;54:1165–72.
- Omland T, de Lemos JA, Sabatine MS, et al. A sensitive cardiac troponin T assay in stable coronary artery disease. N Engl J Med 2009; 361:2538–47.
- 62. de Lemos JA, Drazner MH, Omland T, et al. Association of troponin T detected with a highly sensitive assay and cardiac structure and mortality risk in the general population. JAMA 2010;304: 2503–12.
- Roe MT, Messenger JC, Weintraub WS, et al. Treatments, trends, and outcomes of acute myocardial infarction and percutaneous coronary intervention. J Am Coll Cardiol 2010;56:254–63.
- 64. Williams DO, Holubkov R, Yeh W, et al. Percutaneous coronary intervention in the current era compared with 1985–1986: The national heart, lung, and blood institute registries. Circulation 2000;102:2945.

- 65. Klein LW, Uretsky BF, Chambers C, et al. Quality assessment and improvement in interventional cardiology. A position statement of the society of cardiovascular angiography and interventions. I. Standards for quality assessment and improvement in interventional cardiology. Cath Cardiovasc Int 2011;77:927–35.
- Eggers KM, Lind L, Venge P, Lindahl B. Will the universal definition of myocardial infarction criteria result in an overdiagnosis of myocardial infarction? Am J Cardiol 2009;103:588–91.
- 67. Brandt RR, Filzmaier K, Hanrath P. Circulating cardiac troponin I in acute pericarditis. Am J Cardiol 2001;87:1326–8.
- 68. Smith SC, Ladenson JH, Mason JW, Jaffe AS. Elevations of cardiac troponin I associated with myocarditis. Experimental and clinical correlates. Circulation 1997;95:163–8.
- Boden WE, O'Rourke RA, Teo KK, et al. Optimal medical therapy with or without PCI for stable coronary disease. N Engl J Med 2007; 356:1503–16.
- Bangalore S, Pursnani S, Kumar S, Bagos PG. Percutaneous coronary intervention versus optimal medical therapy for prevention of spontaneous myocardial infarction in subjects with stable ischemic heart disease. Circulation 2013;127:769–81.
- 71. Gustavsson CG, Hansen O, Frennby B. Troponin must be measured before and after PCI to diagnose procedure-related myocardial injury. Scand Cardiovasc J 2004;38:75–9.
- Miller WL, Garratt KN, Burritt MF, et al. Baseline troponin level: Key to understanding the importance of post-PCI troponin elevations. Eur Heart J 2006;27:1061–9.
- 73. Kini A, Marmur JD, Kini S, et al. Creatine kinase-MB elevation after coronary intervention correlates with diffuse atherosclerosis, and lowto-medium level elevation has a benign clinical course: Implications for early discharge after coronary intervention. J Am Coll Cardiol 1999;34: 663-71
- 74. Hong MK, Mehran R, Dangas G, et al. Creatine kinase-MB enzyme elevation following successful saphenous vein graft intervention is associated with late mortality. Circulation 1999;100:2400–5.
- Tardiff BE, Califf RM, Tcheng JE, et al. Clinical outcomes after detection of elevated cardiac enzymes in patients undergoing percutaneous intervention: IMPACT-II. J Am Coll Cardiol 1999;33:88–96.
- Dangas G, Mehran R, Feldman D, et al. Postprocedure creatine kinase-MB elevation and baseline left ventricular dysfunction predict one-year mortality after percutaneous coronary intervention. Am J Cardiol 2002;89:586–9.
- 77. Ellis SG, Chew D, Chan A, et al. Death following creatine kinase-MB elevation after coronary intervention: Identification of an early risk period: Importance of creatine kinase-MB level, completeness of revascularization, ventricular function, and probable benefit of statin therapy. Circulation 2002;106:1205–10.
- 78. Ioannidis JP, Karvouni E, Katritsis DG. Mortality risk conferred by small elevations of creatine kinase-MB isoenzyme after percutaneous coronary intervention. J Am Coll Cardiol 2003;42:1406–11.
- Ajani ÁE, Waksman Ř, Sharma AK, et al. Usefulness of periprocedural creatinine phosphokinase-MB release to predict adverse outcomes after intracoronary radiation therapy for in-stent restenosis. Am J Cardiol 2004;93:313–7.
- 80. Jang JS, Hong MK, Park DW, et al. Impact of periprocedural myonecrosis on clinical events after implantation of drug-eluting stents. Int J Cardiol 2008;129:368–72.
- 81. Andron M, Stables RH, Egred M, et al. Impact of periprocedural creatine kinase-MB isoenzyme release on long-term mortality in contemporary percutaneous coronary intervention. J Invasive Cardiol 2008;20:108–12.
- Garbarz E, Iung B, Lefevre G, et al. Frequency and prognostic value of cardiac troponin I elevation after coronary stenting. Am J Cardiol 1999; 84:515–8.
- 83. Bertinchant JP, Polge A, Ledermann B, et al. Relation of minor cardiac troponin I elevation to late cardiac events after uncomplicated elective

- successful percutaneous transluminal coronary angioplasty for angina pectoris. Am J Cardiol 1999;84:51–7.
- Fuchs S, Kornowski R, Mehran R, et al. Prognostic value of cardiac troponin-I levels following catheter-based coronary interventions. Am J Cardiol 2000;85:1077–82.
- Cantor WJ, Newby LK, Christenson RH, et al. Prognostic significance of elevated troponin I after percutaneous coronary intervention. J Am Coll Cardiol 2002;39:1738–44.
- 86. Herrmann J, Von Birgelen C, Haude M, et al. Prognostic implication of cardiac troponin T increase following stent implantation. Heart 2002;87:549–53.
- 87. Nallamothu BK, Chetcuti S, Mukherjee D, et al. Prognostic implication of troponin I elevation after percutaneous coronary intervention. Am J Cardiol 2003;91:1272–4.
- 88. Kizer JR, Muttrej MR, Matthai WH, et al. Role of cardiac troponin T in the long-term risk stratification of patients undergoing percutaneous coronary intervention. Eur Heart J 2003;24:1314–22.
- Natarajan MK, Kreatsoulas C, Velianou JL, et al. Incidence, predictors, and clinical significance of troponin-I elevation without creatine kinase elevation following percutaneous coronary interventions. Am J Cardiol 2004;93:750–3.
- Kini AS, Lee P, Marmur JD, et al. Correlation of postpercutaneous coronary intervention creatine kinase-MB and troponin I elevation in predicting mid-term mortality. Am J Cardiol 2004;93:18–23.
- Cavallini C, Savonitto S, Violini R, et al. Impact of the elevation of biochemical markers of myocardial damage on long-term mortality after percutaneous coronary intervention: Results of the CK-MB and PCI study. Euro Heart J 2005;26:1494–8.
- Nageh T, Sherwood RA, Harris BM, Thomas MR. Prognostic role of cardiac troponin I after percutaneous coronary intervention in stable coronary disease. Heart 2005;91:1181–5.
- Prasad A, Singh M, Lerman A, et al. Isolated elevation in troponin T after percutaneous coronary intervention is associated with higher longterm mortality. J Am Coll Cardiol 2006;48:1765–70.
- Nienhuis MB, Ottervanger JP, Bilo HJ, et al. Prognostic value of troponin after elective percutaneous coronary intervention: A metaanalysis. Cath Cardiovasc Interv 2008;71:318–24.
- 95. Hubacek J, Basran RS, Shrive FM, et al. Prognostic implications of C-reactive protein and troponin following percutaneous coronary intervention. Can J Cardiol 2009;25:e42–7.
- De Labriolle A, Lemesle G, Bonello L, et al. Prognostic significance of small troponin I rise after a successful elective percutaneous coronary intervention of a native artery. Am J Cardiol 2009;103:639–45.
- Feldman DN, Minutello RM, Bergman G, et al. Relation of troponin I levels following nonemergent percutaneous coronary intervention to short- and long-term outcomes. Am J Cardiol 2009; 104:1210–5.
- 98. Cavallini C, Verdecchia P, Savonitto S, et al. Prognostic value of isolated troponin I elevation after percutaneous coronary intervention. Circ Cardiovasc Interv 2010;3:431–5.
- Feldman DN, Kim L, Rene AG, et al. Prognostic value of cardiac troponin-I or troponin-T elevation following nonemergent percutaneous coronary intervention: A meta-analysis. Cath Cardiovasc Interv 2011;77:1020-30.

Key Words: cardiac biomarkers ■ coronary revascularization ■ myocardial infarction definition.



For supplemental information and tables and citations for references 54 to 99, please see the online version of this article.